

REVIEW HISTORY

Manuscript Title: *Àgó: A Feminine Purification Rite in Òwò, Oṅdó State, South-West, Nigeria*

Submitted: 08 March, 2024

Accepted: 26 June, 2024

Published as ESI Preprint:

Published:

Corresponding Author: Deborah Arowosegbe

Email: deborah.arowosegbe@aau.edu.ng

PEER REVIEW

Reviewer 1: Blinded

Reviewer 2: Blinded